Author's response to reviews

Title: Community-owned resource persons for malaria vector control: enabling factors and challenges in an operational programme in Dar es Salaam, United Republic of Tanzania

Authors:

Prosper P Chaki (pchaki@ihi.or.tz)
Stefan Dongus (sdongus@ihi.or.tz)
Ulrike Fillinger (ulrike.fillinger@lshtm.ac.uk)
Ann Kelly (ann.kelly@lshtm.ac.uk)
Gerry F Killeen (gkilleen@ihi.or.tz)

Version: 4 Date: 26 September 2011

Author's response to reviews:

edit_format